ss16-May-03
Week 21: (19/05 to 23/05)

SM18 Operation SCHEDULE (Monday Morning) Meeting 

· Test Coordinator this week:    O.Berrig

· MTA Technician on call:  (16-3602) G.D’Angelo.

1. Planning for this week : 
· TBA1:   1022  :  First run at 1.9 K
· TBA2:   2006  Test at cold after thermal cycle and warm up: 1016 is coming.
· TBC2 :  3013 at 1.9 K (second run after thermal cycle) 
·  TBC1 : 1024 at 4.2 K tested when 3013 is finished.
· *** CONTINUE  USAGE of “ To-do list ”    *** 
NEW LIST (MASTER COPY) ON PUT On BOARD FOR Thermal Cycle Type Magnets 
2.  Follow-up of problems from Operation Last week( Olav+ Richard)
· HV test at warm or cold. All the cables should be re-connected after the test. Follow the procedures!

· Quench at 1350 mbar at 2 A/s: Always 2 quench at 4.2 K

· Quench at 1.9 K : When AQA is displaying red alarm : stop the tests. (problems with all the quenches this week in 2006 ).

· Do not forget to document the tests: update of the logbook, quench tables, quench history.

-Logbook : operation, problem, actions

-Test sequence : the sequence of the tests (one file per magnet)

-Quench table: Quench results

-Protocol MM : Problems , description of MM.

· Test master on C bench: problems with the MM settings for TBC2. 3013 setting was destroyed.

· MMP: MXI errors on TBC2.

